Author's response to reviews

Title: The association between air travel and deep vein thrombosis: Systematic review & meta-analysis

Authors:

Dr Yaser Y Adi (y.adi@bham.ac.uk)
Sue S Bayliss (s.bayliss@bham.ac.uk)
Andrew A Rouse (Andrew.Rouse@hobtpct.nhs.uk)
Dr Rod R Taylor (r.s.taylor@bham.ac.uk)

Version: 4 Date: 10 May 2004

The following have been addressed as requested:
Formatting changes requested  
-----------------------------------

Title page - Please include email addresses for all authors. (done)

Abstract - Please make sure that this starts on the second page of the manuscript and not on the title page. (done)

Competing interests- Please just write 'None declared'.(done)

Funding - We ask that the funding information is placed in the acknowledgements section of the manuscript. (done)

References - Citations in the reference list should not contain the suffix et al. but list all named authors, regardless of how many there are. (done)